Author’s response to reviews

Title: Contraceptive Use before First Pregnancy by Women in India (2005-2006): Determinants and Differentials

Authors:

Anjali Pandey (anjalionmail@gmail.com)

K Singh (kksingh@bhu.ac.in)

Version: 3 Date: 30 Oct 2015

Author’s response to reviews:

To

The Managing Editor-in-Chief

BMC Public Health

Subject: Submission of revised manuscript in BMC Public Health

Dear Sir,

Attached herewith, please find the revised version of our paper whose manuscript number is PUBH-D-15-00454, for possible publications in your reputed journal BMC Public Health. I have gone through comments of reviewers and immensely thankful for their valuable suggestions. I would like to inform you that the map depicted in figure 1 is my own and has not been taken from another source. Therefore, no permission letter from any other source is required for its use.

Anjali Pandey
Junior Research Fellow
Faculty of Science
Department of Statistics
Banaras Hindu University
Varanasi, India.
Email: anjalionmail@gmail.com